Author's response to reviews

Title: Individual and contextual factors related to dental caries in underprivileged Brazilian adolescents

Authors:

Fabiana L Vazquez (fabilivazquez@gmail.com)
Karine L Cortellazzi (karinecortellazzi@gmail.com)
Armando K Kaieda (armandokaieda@hotmail.com)
Jaqueline V Bulgareli (jaquelineb Bulgareli@gmail.com)
Fábio L Mialhe (mialhe@fop.unicamp.br)
Gláucia MB Ambrosano (glaucia@fop.unicamp.br)
Elaine PS Tagliaferro (epstag@gmail.com)
Luciane M Guerra (lumiranda1302@gmail.com)
Marcelo C Meneghim (meneghim@fop.unicamp.br)
Antonio C Pereira (apereira111@gmail.com)

Version: 5  Date: 24 November 2014

Author's response to reviews: see over
November 24, 2014.

Dear Professor Thiago Machado Ardenghi
Associate Editor BMC Oral Health (Section Epidemiology of Oral Health)

**Paper:** MS: 1572417255127702
**Title:** Multilevel analysis of caries in underprivileged adolescents of Brazilian southeast.

We are resubmitting the referred manuscript to the BMC Oral Health. We have revised it according further editorial revisions:

1. **Please rename 'Objectives' as 'Background' editorial revisions:**
   
   Response: The corrections inside the manuscript are in red color.

2. **Please include a conclusion section in your manuscript, according to our guidelines:**
   
   Response: The corrections inside the manuscript are in red color.

Yours Sincerely

Fabiana Vazquez